Author's response to reviews

Title: Constrictive pericarditis in a post renal transplant subject: A case report

Authors:

Ranga M Weerakkody (rangamw2003@yahoo.com)
Harshani D Perera (harshani1983@gmail.com)
Chaminda H Kularathne (kularathnech@yho.com)
Mohammed HR Sheriff (rezvi.sheriff@gmail.com)

Version: 4 Date: 2 September 2013

Author's response to reviews:

September 2, 2013
Editorial Team,
The Journal of Medical Case Reports,
Dear Sir / Madam,

We have made the suggested changes in the manuscript. Here are the summary of changes we have made.

1. Edited wordings the conclusion section. The heading / sections of the manuscripts are ordered according to the order given on the link http://www.jmedicalcasereports.com/authors/instructions/casereport
2. Figure and Table titles and lists were inserted after the Reference list
3. The edited sections were marked in blue

Looking forwards for a favorable reply

Thank you

Dr. RM Weerakkody